Author's response to reviews

Title: Kinetics of mycolactone in human subcutaneous tissue during antibiotic therapy for Mycobacterium ulcerans disease.

Authors:

Fred S Sarfo (stephensarfo78@gmail.com)
Richard O Phillips (rodamephillips@gmail.com)
Zhang Jihui (jzhang@squ.ac.uk)
Abass K Mohammed (abass@agogopresbyhospital.org)
Abotsi Justice (justiceabotsi@yahoo.com)
Amoako A Yaw (yamoako2002@yahoo.co.uk)
Adu-Sarkodie Yaw (yasax@hotmail.co.uk)
Robinson Clive (c.robinsonsgul@me.com)
Wansbrough-Jones H Mark (wansbrough-jones.2@blueyonder.co.uk)

Version: 4  Date: 27 March 2014

Author's response to reviews: see over
The Editor,
BMC Infectious Diseases

Dear Editor,

We are most grateful for accepting our manuscript for publication in your journal. We were asked to provide primer sequences used in our analysis. A statement “The PCR was based on the IS2404 insertion element; PU4F (5′-GCGCAGATCAACTTCGCGGT-3′) and PU7Rbio (5′-GCCCGATTGGTGCTCGGTCA-3′) were used as primers” has been inserted in the manuscript under the subsection “Diagnostic confirmation of M. ulcerans disease in study subjects”.

Once again on behalf of all my co-authors, I wish to thank you very much and look forward to submitting more manuscripts in the future.

Sincerely,

Dr. Fred Stephen Sarfo (MBChD, FWACP, PhD)